

https://doi.org/10.1093/pnasnexus/pgad432 Advance access publication 14 December 2023 Research Report

Research ReportDiurnal global ocean surface *p*CO₂ and air–sea CO₂ flux reconstructed from spaceborne LiDAR data

Siqi Zhang D^{a,b,c}, Peng Chen D^{a,b,*}, Yongxiang Hu D^d, Zhenhua Zhang D^{a,b}, Cédric Jamet D^e, Xiaomei Lu D^d, Davide Dionisi D^f and Delu Pan^{a,b,c}

^aState Key Laboratory of Satellite Ocean Environment Dynamics, Second Institute of Oceanography, Ministry of Natural Resources, Hangzhou 310012, China ^bSouthern Marine Science and Engineering Guangdong Laboratory (Guangzhou), Nansha District, Guangzhou 511458, China

^cState Key Laboratory of Tropical Oceanography, South China Sea Institute of Oceanology, Chinese Academy of Sciences, Guangzhou 510301, China

^dNational Aeronautics and Space Administration Langley Research Center, Hampton, VA 23681, USA

^eLaboratoire d'Océanologie et de Géosciences (LOG), Université Littoral Côte d'Opale, CNRS, Université Lille, 62930 Wimereux, France

^fInstitute of Marine Sciences (ISMAR), Italian National Research Council (CNR), Rome - Tor Vergata 700185, Italy

*To whom correspondence should be addressed: Email: chenp@sio.org.cn

Edited By: Levi Thompson

Abstract

The ocean absorbs a significant amount of carbon dioxide (CO₂) from the atmosphere, helping regulate Earth's climate. However, our knowledge of ocean CO_2 sink levels remains limited. This research focused on assessing daily changes in ocean CO_2 sink levels and air-sea CO_2 exchange, using a new technique. We used LiDAR technology, which provides continuous measurements during day and night, to estimate global ocean CO_2 absorption over 23 years. Our model successfully reproduced sea surface partial pressure of CO_2 data. The results suggest the total amount of CO_2 absorbed by oceans is higher at night than during the day. This difference arises from a combination of factors like temperatures, winds, photosynthesis, and respiration. Understanding these daily fluctuations can improve predictions of ocean CO_2 uptake. It may also help explain why current carbon budget calculations are not fully balanced—an issue scientists have grappled with. Overall, this pioneering study highlights the value of LiDAR's unique day–night ocean data coverage. The findings advance knowledge of ocean carbon cycles and their role in climate regulation. They underscore the need to incorporate day–night variability when assessing the ocean's carbon sink capacity.

Keywords: diurnal variation, air-sea CO₂ flux, CALIPSO, LiDAR, remote sensing

Significance Statement

It is very difficult (probably impossible) to measure ocean surface pCO_2 directly from space. The best proxy of pCO_2 that can be measured from space is likely ocean biological properties (e.g. primary productivity, Chla, etc.) that are very sensitive to pH level and thus dissolved organic carbon (DIC)/ pCO_2 , together with other measurable physical properties (e.g. temperature, ocean dynamics, etc.). Training the biological and physical properties of pCO_2 from in situ measurements sounds like a very rational approach. Diurnal changes in ocean surface pCO_2 could affect ocean carbon sinks, which is only beginning to be assessed nowadays. This study helps fill this knowledge gap by employing a cutting-edge satellite technology—LiDAR. LiDAR can probe the oceans day and night using laser pulses. We have developed a new model based on a feed-forward neural network incorporating LiDAR data as input to estimate global ocean CO_2 absorption from 1998 to 2020 at a monthly resolution. For the first time, this approach harnesses unique advantages of LiDAR's day–night ocean observations. The results offer new insights into daily CO_2 fluctuations and informing policy decisions.

Introduction

The global ocean plays a vital role in mitigating climate change by absorbing atmospheric carbon dioxide (CO₂), a heat-trapping greenhouse gas. Scientists estimate that oceans currently absorb about 25% of human-generated CO₂ emissions (1). However, gaps remain in understanding the ocean carbon cycle. Comprehensive global measurements of the surface partial pressure of CO₂ (pCO₂)

are lacking. This hampers efforts to predict ocean CO₂ uptake, model climate change impacts, and inform policymaking.

Comprehending the onset and progression of ocean acidification necessitates a thorough understanding of the entire cycle of CO_2 variability (2). Since 1850, the ocean carbon sink has increased in tandem with the exponential rise in anthropogenic emissions, resulting in the marine CO_2 sink reaching 3.0 ± 0.4 GtC year⁻¹ in

Competing Interest: The authors declare no competing interest. Received: November 1, 2022. Accepted: December 4, 2023



© The Author(s) 2023. Published by Oxford University Press on behalf of National Academy of Sciences. This is an Open Access article distributed under the terms of the Creative Commons Attribution License (https://creativecommons.org/licenses/by/4.0/), which permits unrestricted reuse, distribution, and reproduction in any medium, provided the original work is properly cited.

2020 (3). However, the carbon budget imbalance, which represents the disparity between the estimated total emissions and the estimated changes in the atmosphere, ocean, and terrestrial biosphere, is approximately 0.1–0.3 GtC year⁻¹ (3–5). This imbalance serves as an indicator of incomplete data and our current understanding of the contemporary carbon cycle (4). The long-term variations in the ocean surface pCO₂ act as the principal driving force governing CO_2 exchange across the air-sea interface (6). Consequently, sea surface pCO₂ observations assume a pivotal role in constraining the global air-sea carbon sink. Regrettably, such knowledge remains limited due to the scarcity of comprehensive global sea surface pCO₂ observations. Recent global estimates still obscure noteworthy interannual and regional fluctuations (7), underscoring the necessity for rigorous quantitative research to meticulously trace the precise evolution of the Earth's carbon budget (8).

Benefiting from the growing abundance of global ocean biogeochemical remote sensing data and the continuous advancements in remote sensing retrieval methods, the estimation of long-term series of sea surface pCO_2 is progressively achieving higher levels of accuracy. A variety of data-interpolation approaches provide estimates of the surface ocean pCO_2 field (7) such as statistical interpolation, linear and nonlinear regression, or model-based regressions or tuning (9). Artificial neural networks have succeeded in filling the spatial and temporal gaps. Different artificial neural networks have been widely used to reconstruct sea surface pCO_2 in the global ocean (10–18). In contrast, the existing products usually present monthly fields with a 1° × 1° spatial resolution, with large errors occurring in Antarctica and the Arctic due to the problem with input remote sensing data (18).

While average states and seasonal variations in ocean CO₂ system variables have received relatively comprehensive characterization, the assessment of diurnal changes is only in its infant stages (19, 20). Previous observation-based studies have explored the diurnal variability of surface ocean pCO₂ and reported extreme diurnal amplitudes of $187 \pm 85 \mu$ atm in the open ocean (20), 5–25 µatm in the open-ocean Sargasso Sea (21), 8 and 15 µatm in the tropics and <5 µatm in the subtropics (22). Higher diurnal amplitudes for pCO_2 (100–500 µatm) have been recorded near benthic ecosystems such as coral reefs (20, 23, 24), kelp forests (25), and seagrasses (26), particularly in shallow waters, where the percentage of benthic biomass is enhanced (27, 28). Diel pCO_2 variations provide essential insights into the dynamic processes of the carbon cycle and affect ocean acidification (29). Diurnal pCO₂ variations also influence marine biological productivity, which, in turn, can impact the entire marine food web (30). Moreover, changes in ocean temperature and circulation can alter CO₂ uptake and release processes, potentially leading to positive or negative feedback loops that either amplify or mitigate climate change (31, 32). In summary, understanding of oceanic processes and aids in the formulation of effective environmental policies and conservation measures. It provides valuable data for policymakers and researchers working toward sustainable ocean management and mitigating the impacts of climate change. However, there is no study on global diurnal ocean surface pCO₂ and air-sea CO₂ flux (C-flux) reconstruction considering both diurnal thermodynamic influences and biochemical effects. Moreover, traditional passive remote sensing data failed to provide information during nighttime, adding an additional challenge to addressing this critical knowledge gap.

During the last two decades, the Cloud-Aerosol Lidar with Orthogonal Polarization (CALIOP) has been the primary instrument on the cloud-aerosol LiDAR flown aboard the Cloud-Aerosol Lidar and Infrared Pathfinder Satellite Observations (CALIPSO) platform, providing day-and-night measurements comparable with those of mature passive remote sensing radiometry since June 2006 (33–38). CALIOP is a polar orbiting sensor that conducts daytime and nighttime (approximately 13:40 and 01:40 local time, respectively) nearnadir backscattering measurements along its orbit track at a sampling frequency that is equivalent to every 330 m on the ground. Recent studies have demonstrated that CALIOP also collects information about the ocean at both the global scale and in specific regions: Hu et al. (39) estimated wind speed and wave slope variance on a global scale; Hu et al. (40) suggested that the higher spatial resolution (70 m) wind from CALIOP could help reduce uncertainties in air-sea exchange; Behrenfeld et al. (34) provided global maps of particulate backscattering observed from CALIOP; Lu et al. (35) showed the global sea surface chlorophyll-a concentration (Chla) and particulate organic carbon concentrations estimated from CALIOP; and Behrenfeld et al. (37, 41) measured the annual cycles of phytoplankton biomass in polar regions and studied global satellite-observed daily vertical migrations of ocean animals. Compared with traditional satellite ocean color remote sensing, CALIOP measurements could provide new observations of seawater phytoplankton properties for both day and night, globally and in polar regions, to improve the understanding of global phytoplankton primary productivity and carbon stocks/ fluxes (42-45). Despite the likely importance of these changes, the consequences of increased diel chemistry variation for marine organisms and ecosystem processes remain almost entirely unexplored.

Overall, satellites have enabled remote sensing of key ocean properties related to CO₂ absorption like temperature, plankton levels, and wind speeds. However, traditional satellites only provide data during daylight hours. Nighttime changes in CO₂ linked to winds, temperatures, and biology remain poorly characterized. This study helps fill this knowledge gap by employing a cutting-edge satellite technology-LiDAR. LiDAR can probe the oceans day and night using laser pulses. Within this frame, our objectives were 2-fold: firstly, to explore the potential application of CALIOP diurnal Chla and wind speed products in studying the global air-sea carbon cycle and secondly, to examine the longterm diurnal pCO₂ and air-sea C-flux using multisource satellite products, which include LiDAR products. To achieve this, we have proposed a novel feed-forward neural network model with LiDAR data inputs (FNN-LID) to estimate global long-term diurnal pCO_2 and C-flux. The best proxy for pCO_2 that can be measured from space is likely ocean biological properties (e.g. primary productivity, Chla, etc.) that are very sensitive to pH level and thus DIC/pCO₂, together with other measurable physical properties (e.g. temperature, ocean dynamics, etc.). By employing this model, we were able to establish a nonlinear and continuous relationship between climatology pCO2 and various independent environmental predictors, such as Chla, sea surface temperature (SST), and others, derived from remote sensing data. The study period covered from January 1998 to December 2020 at a monthly 1°×1° resolution. Notably, this research represents the pioneering effort to incorporate spaceborne LiDAR diurnal data in constructing global diurnal ocean surface pCO2 and C-flux. To assess the performance and accuracy of our model, we conducted a comprehensive comparison with eight other estimates based on the fugacity of CO₂ (fCO₂) from Global Carbon Budget 2020 (GCP2020). In particular, we compared the CALIOP-retrieved Chla against consistent observations obtained from biogeochemical Argo profiling floats (46). Similarly, we validated the wind speed data from buoys located across the equatorial and subequatorial oceans.

Zhang et al. | 3

Subsequently, we were able to estimate the long-term series of global ocean surface pCO_2 and C-flux. In addition, we summarize the developed approach and the main results and provide recommendations for future biogeochemical studies using LiDAR active remote sensing measurements. These findings shed light on crucial aspects of the air-sea carbon cycle and contribute to our understanding of the Earth's carbon dynamics on a global scale. In Materials and methods section, we describe the CALIOP data, in situ observation data and other used remote sensing environmental data. Then, we introduce the adopted and refined procedures to retrieve the diurnal Chla and wind speed from CALIOP and the ocean surface pCO_2 and air-sea C-flux estimated based on the FNN-LID method.

Results FNN-LID validation

In this study, we validated the global estimates of FNN-LID pCO₂ data against both observed data and widely used reconstructed datasets. The observed data utilized in the validation process consisted of unmodeled gridded data from the SOCATv2022 Gridded Dataset (47). The results revealed exceptional long-term agreement with a remarkable r^2 value of 0.79, a low root mean square error (RMSE) of 17.74 µatm, and an almost negligible overall bias of 0.05 µatm considering 250,100 matched gridded observations (Fig. 1A). For statistics, please refer to the SI Appendix. This strong global fit, encompassing both open ocean and coastal regions, was consistently observed for each individual year, demonstrating high levels of consistency, with r^2 values ranging from 0.73 to 0.90, RMSE ranging from 15 to 21 µatm and annual bias remaining within a narrow range of ±2 µatm (SI Appendix, Table S1). The reanalysis datasets included pCO2 data reconstructed using FNN from the Copernicus Marine Environment Monitoring Service (CMEMS) and pCO₂ data reconstructed using SOM-FNN from Biogeochemistry and Pollutant Dynamics (IBP) (16, 17, 48). The RMSE for the CMEMS global estimates ranged from 17 to 26 µatm, while the IBP estimates exhibited an RMSE ranging from 9 to 13 μ atm over the open ocean and 4 to 32.9 μ atm over coastal seas. Globally, the FNN-LID model demonstrated an excellent fit with the gridded pCO₂ data from SOCATv2022, showcasing accuracy levels similar to those of other recent models in the mean monthly result. These findings further support the reliability and effectiveness of the FNN-LID approach for global pCO₂ estimation.

The right panel of Fig. 1B presents the temporal mean residuals when comparing them to the SOCAT map in each pixel. Overall, the bias was minimal and exhibited a random distribution in space in most midlatitude open oceans (e.g. North Pacific and North Atlantic). However, it tended to increase in regions sparse or limited observation data (e.g. Indian Ocean and Southern Ocean) as well as in areas characterized by extremely high or low pCO_2 (e.g. East Equatorial Pacific and Labrador Sea). Regions with high spatial variability exhibited relatively poorer model fits, whereas the FNN-LID model demonstrated notably good performance in capturing the dynamics of most less variable open-ocean regions.

To validate the surface ocean pCO_2 across different regional oceans and distinguish the coastal from the open ocean, we utilized the global ocean masks from the World Ocean Database, NOAA (available at https://www.nodc.noaa.gov/OC5/WOD/wod_ masks.html). Further details regarding the regional (open and coastal) division are provided in Fig. 1C. The North Pacific stands out as a region with high data coverage and a rapid increase in data availability since 2003 was observed. The corresponding RMSEs for the North Pacific, the Equatorial Pacific, and the South Pacific were below 18 µatm, and r^2 values ranged between 0.76 and 0.84. Although the RMSEs were generally low across the Pacific, skillful data reconstruction over the coastal North Pacific regions posed challenges. RMSEs were generally below 20 µatm, with a larger RMSE of 39.52 µatm obtained for the Coastal North Pacific. The large model-data mismatch along the Pacific continental shelves reflects the poor reconstruction of pCO₂ over regions under the influence of upwelling systems (e.g. Kuroshio Current), large river discharges (e.g. Bering Sea shelf), and the bottlenecks of gulfs or bays (e.g. South China Sea). As discussed in Hales et al. (49), the carbon cycling in the Coastal Pacific is subject to complex dynamics, exhibiting high spatial and temporal variability driven by multiple physical and biogeochemical drivers. These intricate spatial and temporal changes occur within the 1°×1° grid contribute to the large RMSE for the reconstructed product. On the other hand, a comparison between the gridded observed data and evaluation data across the six subregions of the Atlantic reveals small mean model-data differences, leading to high reconstruction skill in the Atlantic basin (SI Appendix, Table S2). The coastal Atlantic regions performed particularly well in terms of reconstruction, accounting for over 29% of the total coastal data. Mean RMSE is below 25 µatm, and, with the exception of the coastal $(r^2 = 0.71)$. Corresponding RMSDs are 14.33 (North Atlantic), 15.09 (Equatorial Atlantic), and 13.73 µatm (South Atlantic), with r^2 values ranging approximately between 0.66 and 0.77.

In polar regions, FNN-LID pCO_2 exhibits high quality in the Arctic (RMSE of 25.3 µatm), which is comparable with the recent SOCAT-based global pCO_2 fields (RMSEs of 26.7–32 µatm) and provides a more accurate estimate of the polar continental shelf in winter (50). This advantage stems from the fact that FNN-LID uses CALIPSO which has advantages over traditional passive remote sensing (the Sea-viewing Wide Field-of-view Sensor [SeaWiFS] or Moderate Resolution Imaging Spectroradiometer [MODIS]) in distinguishing clouds and sea ice (51). Therefore, active satellite LiDAR input enables continuous observation of the winter polar nighttime period, obtaining high-coverage polar inputs (52). Thus, we constructed a LiDAR-based pCO_2 with far greater coverage than previous products.

In conclusion, the residual analysis demonstrates that the global FNN-LID method successfully meets most criteria for a robust fit, with no significant evidence of hidden biases. Notably, the estimates do not show any substantial degradation concerning data density, be it in temporal or spatial dimensions. Regions with pronounced spatial or temporal variability exhibit relatively weaker fits, whereas the method excels in accurately estimating pCO_2 for most open-ocean regions characterized by lower variability.

Interannual pCO₂ and air-sea flux

Utilizing the FNN-LID methodology, we successfully obtained the climatological surface ocean pCO_2 during 1998–2020 (Fig. 2A). The highest long-term mean sea surface pCO_2 climatology values identified by our FNN-LID method occurred in the equatorial Pacific, which was associated with the strong upwelling of deep water with naturally rich dissolved inorganic carbon. Similarly, elevated pCO_2 levels were also identified in the northeastern Indian Ocean, the low-latitude South Atlantic and the western basin of the Bering Sea (Fig. 2A). Conversely, regions with the lowest sea surface pCO_2 values were detected in the high latitudes (>70°) of the Atlantic, along the strong western boundary current (Gulf Stream and Kuroshio Current), and in the subtropical bands of the Southern Hemisphere, a combination of both the drawdown



Fig. 1. A) Observation of global surface ocean pCO_2 (µatm) and FNN-LID estimates. The color code represents the density of occurrences. The line is the 1:1 line. B) Map of the mean of the differences between FNN-LID pCO_2 and SOCAT gridded dataset (47). The surface ocean pCO_2 is colocalized with observations and the bias is calculated. Then, this bias is summed in time and the average. C) The masks of 26 global oceans from World Ocean Database, NOAA to validate the surface ocean pCO_2 in different regional oceans (available at https://www.nodc.noaa.gov/OC5/WOD/wod_masks.html).

of DIC by biological activity and the low-temperature effect on solubility (12). Furthermore, we conducted an analysis of the interannual increasing rate for each 1°×1° pixel, as illustrated in Fig. 2B. The results indicate a significant upward trend, with an average increasing rate of approximately 1.8 ($\pm 0.7 \mu$ atm year⁻¹), among which the Ross Sea, East Siberian Sea, and Beaufort Sea growth was set to be sharply over 4 μ atm year⁻¹. These findings were corroborated by multiple models, and to validate our approach, we compared the global monthly average pCO2 data with CMEMS and the Institute of IBP over the period from 1998 to 2020, as presented in Fig. 2C. The comparison of sea surface pCO₂ from the three mapping approaches displayed minor variations within the range of ± 3 µatm. On a global scale, the three products gave extremely similar results for a long time series. However, it is worth noting that the IBP method exhibited a slight overestimation of observed pCO₂, particularly during the period 1998–2003, with deviations of up to 3 µatm. On the other hand, the discrepancy between the CMEMS output and our approach fluctuated within the range of $\pm 2 \mu$ atm, with an increase in amplitude of up to -4 µatm from November 2020.

Figure 2D displays the long-term mean C-flux density in mol C m⁻² year⁻¹, providing insights into the average annual net atmosphere–ocean C-flux. The results indicated CO₂ release predominantly at low latitudes and uptake at high latitudes, with the exception of the Bering Sea, which acts as a source of CO₂ to the atmosphere. The detailed calculation steps for the C-flux are available in SI Appendix. Remarkably, we identified significant sink areas in the transition zone between the subtropical gyre and subpolar waters (i.e. 30°–50° latitudes). This phenomenon primarily arises due to the cooling of subtropical warm waters and secondarily to the biological drawdown of pCO_2 in nutrient-rich subpolar waters. Moreover, areas with high wind speeds were found to enhance the CO_2 sink capability of ocean waters. The eastern equatorial Pacific, northwestern Arabian Sea and western Bering Sea were the most intense CO_2 source areas (red regions in Fig. 2D). The equatorial Pacific, particularly in the eastern upwelling area, was highly supersaturated with regard to atmospheric pCO_2 and constituted a strong source region for air. Additionally, the tropical Atlantic, Pacific, and Indian Oceans and the subarctic region, also exhibited significant source characteristics for CO_2 .

The regional distribution of C-flux in the world's oceans has been a subject of scientific interest, and recent research highlights the significance of previously overlooked areas, such as the Arctic. In the past, limited data coverage of input data in previous studies has taken up approximately 15+% of the global ocean CO₂ uptake flux (SI Appendix, Table S3). Similarly, the Atlantic Ocean covers roughly 23% of the ocean area and accounts for approximately 31+% of the global ocean CO_2 uptake. In contrast, the Pacific Ocean accounts for only 25% of the global ocean CO₂ uptake, whereas it occupies 47% of the global ocean area. The lower CO₂ sink is formed by the juxtaposition of the intense CO₂ source in the equatorial Pacific with a strong seasonal source in the Bering Sea. The dynamics of this region are further influenced by El Niño events, involving decreased upwelling of carbon in the equatorial Pacific due to a weakening of the trade winds. This region will become a weaker sink of CO₂ or will become near neutral if the El Niño event is strong (53). The total CO_2 flux was scaled by the ratio of the total ocean area covered by the respective product to the total ocean area $(361.9 \times 10^6 \text{ km}^2)$ (54, 55). Furthermore, it should be noted that the ocean sink we discussed here does not contain carbon from river inputs to the ocean, which are approximately 0.61 GtC year⁻¹ (the average of 0.45 ± 0.18 GtC year⁻¹ by



Fig. 2. The distribution of A) climatological annual mean pCO_2 in μ atm and B) mean growth rate in μ atm year⁻¹. C) Temporal evolution of the global pCO_2 (in μ atm) from blue FNN-LID, IBP, CMEMS, and estimates during 1998–2020. D) Map of the long-term mean annual net air–sea flux for CO_2 (mol C m⁻² year⁻¹) for 1998–2020. Red–yellow areas indicate that the ocean is a source for atmospheric CO_2 , and blue–indigo areas indicate that the ocean is a CO_2 sink. E) Comparison of the global carbon budget in this study (black line) with the 8 fCO_2 -based estimates from GCP2020 from 2001 through 2020. All values are in GtC year⁻¹, and positive flux densities indicate CO_2 uptake by the ocean.

(56) and 0.78 ± 0.41 GtC year⁻¹ by (57)). Figure 2E illustrates the result that our FNN-LID-based global contemporary air–sea C-flux exhibited a modest level of year-to-year variability from 2001 to 2020, with a minimum carbon uptake of -1.10 ± 0.32 PgC year⁻¹ in 2001 and reached a maximum uptake of -2.7 ± 0.47 PgC year⁻¹ in 2020 and with a standard deviation of the deseasonalized and detrended (to separate the effect of short-term trends) monthly fluxes of (± 0.09 PgC year⁻¹). We showed a similar result in both annual estimation and interannual variation to the eight results mentioned in GCP2020 (3).

The global diurnal ocean surface pCO₂ and flux

The above sections have substantiated the availability of LiDAR products for their potential application in pCO_2 studies. Furthermore, we have demonstrated diurnal pCO_2 based on CALIPSO satellite products and multisource diurnal products. We used the diurnal SST and Chla to reconstruct the diurnal pCO_2 fields (henceforth referred to as the FNN-LID) by applying the FNN-LID, and the climatological diurnal differences in SST and Chla are shown in Fig. 3A and B. Fields of diurnal and nocturnal transfer velocity were computed using wind speed observations from CALIPSO. These fields were then used to compute the monthly mean fields of the daytime and nighttime air-sea CO_2

fluxes. Due to the resolution of the $1^{\circ}\times1^{\circ}$ grid and the lack of diurnal temperature data at high latitudes, our credible diurnal pCO_2 data would cover open oceans between latitudes of 60°S-60°N, as shown in Fig. 3C. Next, we specifically describe the influencing factors and potential causes of diurnal variation in seawater pCO_2 .

Temperature is a pivotal parameter in carbonate systems because it affects pCO_2 in isochemical conditions ($\partial \ln pCO_2/\partial T$) with a rate of +4.23% °C⁻¹ (58). The acute daily cycle is influenced by solar radiation, wind speed, the optical attenuation coefficient of the water, and mixing from the wave motions (59). This diurnally varying SST rather than daily averaged SST affects air–sea fluxes (60). Figure 3A indicates that the mean amplitude in Δ SST for the entire ocean is 0.47°C (±0.26°C), with the largest Δ SSTs exceeding 1.7°C in Indonesia and affecting 0.01% of the surface.

However, the thermodynamic effect was dampened by primary production during the day, as shown in Fig. 3B. Light affects phytoplankton activity, leading to diel periodicity in cell division and cellular properties (61–66). The Chla difference ratio (Eq. 1) yields a negative value in almost the entire open sea (Fig. 3B):

$$\Delta \text{Chla} = \frac{\text{Chla}_{\text{day}} - \text{Chla}_{\text{night}}}{\text{Chla}_{\text{day}}} \times 100\% \tag{1}$$

In contrast, the marginal sea Chla concentrations were higher during the day, such as in the northwestern Arabian Sea, Gulf of



Fig. 3. The annual mean difference of day and night A) SST, B) Δ Chla, and C) pCO₂, among Δ Chla is shown as the normalized difference ratio. C) The map of global diurnal air–sea C-flux difference (day minus night) in mol m⁻² year⁻¹; D) comparison of the CO₂ flux showing the budget values of ocean at nighttime and at daytime in GtC year⁻¹.

California, and the Arafura Sea. The same pattern was found in equatorial regions, especially the eastern equatorial Pacific. The global annual mean diurnal variation pattern of Δ Chla agreed well with the study by Behrenfeld et al. (41). Synchronized cell division during nighttime has been confirmed for many phytoplankton groups, resulting in a decrease in cell size during this period of the day, which may cause diel changes (41, 61, 63, 65, 66). Regardless of the ecological reasons behind the existence of diel rhythms in Chla, their consequences in day and night are different. In addition, these diel variations have been neglected in all previous research regarding the dependence of passive remote sensing upon solar radiation.

The annual mean diurnal variations in pCO_2 (d pCO_2 = daynight) are displayed in Fig. 3C. The variations are a comprehensive consequence of diurnal SST variations, leading to an increase (due to deep-water upwelling and sea-air CO₂ exchange) or decrease (due to photosynthesis and sea-air CO₂ exchange) in the total CO₂ concentration. The spatial distribution of these variations is unique, showing positive $dpCO_2$ in the subequatorial and midlatitude regions (higher diurnal pCO_2), while it was negative in the equatorial Pacific and Southeast Pacific (higher nocturnal pCO_2). The lowest $dpCO_2$ (>10 µatm) was exhibited in Patagonia. Diurnal variations were small in the tropics and typically <5 µatm. The strongest diurnal variations occurred in the northern subtropics and midlatitudes, where we estimated that pCO_2 was typically lowered by between 3 and 7 µatm during nighttime compared with daytime. In the Southern Hemisphere, diurnal variations were lower, between 3 and 6 μ atm. Negative dpCO₂ appeared in upwelling regions, such as in the western Arabian Sea. Many previous studies have shown that an increase in seawater temperature will cause changes in the carbonate composition of seawater, resulting in a larger pCO₂. In summary, diurnal pCO₂ change is not a simple change due to temperature or Chla but is rather caused by a combination of seawater inorganic carbonate systems and a series of complex organic biochemical processes.

As for diurnal flux, there are differences in the magnitude of the fluxes between day and night (Fig. 3D). The diurnal flux variations are mostly positive, indicating that the evasion in source regions is increased during the day, whereas the invasion in sink regions is decreased. Diurnal flux variations are strongest in the southern midlatitudes where the day-night flux difference is ~1 mol m^{-2} year⁻¹. However, in the tropics and West Arabian Sea diurnal variations are negative and sometimes even <0.5 mol m⁻² year⁻¹ (e.g. at 100°W). Combined with the climatology air-sea C-flux, the regions that be deemed as carbon sources are more likely to release more CO₂ at night than during the day. Figure 3E shows the long-term series global flux, revealing that the nocturnal sink is higher than the diurnal one. Furthermore, if we keep using the diurnal uptake as the whole daily sinks, it will cause underestimation by between -0.005 and 0.052 GtC year⁻¹ (0.026 GtC year⁻¹ on average). This may be helpful to answer the question of where part (approximately 10-30%) of the total carbon budget imbalance (about 0.1-0.3 GtC year⁻¹) comes from, over the last decade.

Diurnal pCO₂ plays a crucial role in unraveling the complexities of the carbon cycle, ocean acidification, biological productivity, climate change feedbacks, and environmental health. Jury et al. (29) showed that under global change diel seawater chemistry variation increases (dramatically in some cases) and that various ecosystem feedbacks can substantially modify changes in both the average chemistry and diel chemistry variation. Delille et al. (67) studied diel fluctuations of pCO₂ and DIC inside and outside a giant kelp bed and suggested that understanding the physical and biological processes regulating pCO₂ dynamics facilitates tracking the seasonal evolution of primary production. A workshop on the potential impacts of ocean acidification on marine ecosystems and fisheries indicated the precise control of diel CO₂ cycling was considered highly valuable in studying the impact of ocean acidification (30). However, diurnal change in ocean surface pCO₂ could affect ocean sink, which is only beginning to be assessed nowadays. This study on diurnal variation may be helpful in answering the question where some of the global carbon budget imbalance over the past decade comes from. Its importance extends to various scientific disciplines, helping shape our understanding of the Earth's interconnected systems and informing strategies to safeguard marine ecosystems and global environmental stability.

Discussion

Global seasonal amplitude

The seasonal variations in sea surface pCO_2 , as observed at the Hawaiian Ocean time-series station (68) (1.5 ± 1.8 and 0.2 ± 1.8 µatm per decade) and at the Hydrostation "S"/Bermuda Atlantic time-series study site (69–71) (1.5 ± 1.1 µatm per decade) showed a significantly increased in recent decades. Meanwhile, model-based projections supported this result, especially in the Southern, Pacific, and North Atlantic Oceans (2, 72, 73).

Based on the FNN-LID methods, we reconstructed changes in global ocean pCO_2 on a monthly basis with a spatial resolution of $1^{\circ} \times 1^{\circ}$ for the period 2001 to 2020. To assess seasonal differences, we calculated winter averages in the Northern Hemisphere as the mean of December, January, and February and the summer averages as the mean of June, July, and August; and the reverse was applied for the Southern Hemisphere. The seasonal differences in pCO_2 were computed here as summer averages minus winter averages. This approximation slightly underestimated seasonal oscillations in the equatorial region, where the highest value may not occur in the summer due to seasonal changes in solar radiation.

Analyzing the mean result over 5-year intervals, we observed that the seasonal differences in surface pCO_2 increased in the Northern Hemisphere, with the winter-to-summer differences becoming more positive in low latitudes (equatorward of ~40°) and more negative in high latitudes (poleward of ~40°) (SI Appendix, Fig. S1A–D). Similar spatial patterns were noted in the Southern Hemisphere with a slightly increasing summer-minus-winter difference, except in the South Pacific (SI Appendix, Fig. S1E and F). The change in sign over 40° latitude corresponded to a 6-month phase shift in the seasonal peak of pCO_2 being 6 months out of phase between these two bands. In high latitudes, the seasonal cycle had a maximum in summer, resulting in a negative seasonal difference. In contrast, the seasonal cycle in low latitudes exhibited a peak in winter, leading to a positive summer-minus-winter difference in pCO_2 .

Although some residual interannual variability persisted in the strength of the seasonal pCO_2 difference, it is essential to

emphasize the explicit positive trend in northern extratropical regions, with an average rate of 2.1 \pm 0.6 $\,\mu atm$ per decade.

Accuracy of diurnal air–sea C-flux

Errors in the reconstruction of the global diurnal pCO₂ arise from both the FNN-LID methods and the input data. Unlike the monthly average product, here, our input data were newly added, including diurnal SST, sea surface Chla, wind speeds, and atmospheric pCO₂.

The daily and monthly night SSTs were handled from longwave infrared (LWIR) SST products, a 30+ year record of space-based measurements of LWIR SST from polar orbiting satellites. The current LWIR algorithm is applicable for both daytime and nighttime observations and is based on a modified version of the nonlinear SST algorithm developed by Walton et al. (74), most recently described in Kilpatrick et al. (75). The SST measured by MODIS and VIIRS infrared radiometers is commonly referred to as the skin temperature of the ocean and not the body of water below, as measured by in situ thermometers (76). The thermal skin layer of the ocean is <1 mm thick and cooler than the underlying water (77, 78). For the uncertainty validation for the ESA analysis dataset, the spread of uncertainties in the products ranged from 0.05 to 1.5 K, and the agreement between the theoretical and measured RMSE values was excellent across the full range of uncertainties (79).

For Chla, the global distribution of ocean subsurface backscatter was estimated from the CALIOP level-1 data product, retrieving a diurnal Chla concentration with an accuracy that is (at least as good as) comparable with that obtained from MODIS (35), which can help us better understand the different optical properties of various water masses and facilitate the quantification of global ocean carbon stocks (80). In our analysis, we used MODIS Chla rather than GlobColour Chla, due to the merged product is merging different errors (such as seasonal bias) and uncertainties and the result is not as robust as using one satellite with the same time period as CALIOP (81). The comparison between the independent validation MODIS dataset and the daytime Chla-LID used in this study is shown in SI Appendix, Fig. S2A with an RMSE of 0.85 μ g/L and an r^2 value of 0.75. To estimate the accuracy of the nocturnal data, we also compared our products with BIO-Argo within 1° × 1° and 6 h. Although there is a difference between the Chla-LID and BIO-Argo Chla, the bias is insensitive to whether the measurement is taken during the daytime or nighttime, as shown in SI Appendix, Fig. S2B. Although the comparison statistics results for the BIO-Argo measurement were lower than the former, our day and night data maintained a similar estimation ability, providing high-quality nighttime data that could not be retrieved from MODIS. Roesler et al. (82) found that Argo-Chla can be biased by a factor of 2. Additionally, other biological impacts, such as zooplankton diurnal changes, may explain the bias between Argo and CALIOP.

The diurnal wind speed utilized in our study was retrieved from CALIPSO. To validate the neural network approach, we applied CALIPSO measurements for the entire dataset from 2008. The retrieval results were unbiased, with an average wind speed difference between CALIPSO and AMSR-E of approximately -0.32 m/s (SI Appendix, Fig. S2C). Meanwhile, we selected measured buoy data from the National Data Buoy Center, PIPATA, RAMA, and TAO/TRITON as the standards to verify the wind speed. The comparison indicated that the CALIPSO wind speeds may be underestimated (by 1–2 m/s) when the wind speed is <8 m/s in tropical regions (SI Appendix, Fig. S2D); Nevertheless, the CALIPSO data remains globally applicable, particularly for capturing instantaneous wind speeds. Due to the gradient of wind speed with

latitude, although the wind speed model we have established has a good inversion capability on a global scale, but some errors may occur in specific regions (e.g. underestimation of low wind speeds in low latitudes). The relationship between mean square wave slope (LiDAR "wind speed") and air-sea turbulence exchange speed is independent of stability, but the relation between the true wind speed (and thus microwave-based wind measurements, which is a simple empirical fit of buoy-measured wind) and turbulence exchange speed is stability dependent. LiDAR is actually measuring mean square slopes (or turbulence exchange speed) directly and thus its "wind speed" is the correct wind speed for neutral stability. It is a little bit trickier for microwave (radar or radiometer) measurements, since the microwave wavelength (cm) is too close to the wavelength of capillary-gravity waves (cm too). Nighttime mean square slopes (which LiDAR measures directly) might be larger than the daytime ones even if the wind speeds of day and night are the same. So, LiDAR is actually measuring the diurnal change of the turbulence exchange speed. And the diurnal change in turbulence exchange may be underestimated if we use wind derived from microwave measurements.

Effect of the diurnal air-sea C-flux variation

We used the diurnal pCO_2 fields of the FNN-LID and the fields of diurnal xCO_2 and wind speed to estimate the monthly mean fields of daytime and nighttime air-sea C-flux. The calculation of air-sea C-flux appeared in the form of multiplication in this study. Therefore, we will discuss the impact of different mechanisms on C-flux in terms of day-night ratios below:

$$\frac{\text{flux}_{\text{day}}}{\text{flux}_{\text{night}}} = \frac{\text{sol}_{\text{day}}}{\text{sol}_{\text{night}}} \times \frac{kw_{\text{day}}}{kw_{\text{night}}} \times \frac{\Delta p_{\text{day}}}{\Delta p_{\text{night}}} = k_{\text{sol}} \times k_{kw} \times k_{\Delta p}$$
(2)

$$k_{\rm sol} = \frac{\rm sol_{day}}{\rm sol_{night}} \tag{3}$$

$$k_{kw} = \frac{kw_{day}}{kw_{night}} \tag{4}$$

$$k_{\Delta p} = \frac{\Delta p_{\text{day}}}{\Delta p_{\text{night}}} \tag{5}$$

where flux_{day}, sol_{day}, kw_{day} , and Δp_{day} represent the air–sea CO₂ flux, the solubility of CO₂ in seawater, the gas transfer velocity, and the difference in CO₂ partial pressure between the ocean surface and the atmosphere during the day; while flux_{night}, sol_{night}, kw_{night} , and Δp_{night} for the night. Furthermore, k_{sol} , k_{kw} , and $k_{\Delta p}$ characterize the ratios of the different components in the day and night, respectively (Fig. 4A–C). When this ratio is <1, it means that the parameter is lower during the day, and the opposite means that the larger value occurs during the day.

In previous studies, the combined effect of $k_{\rm sol}$ was found to be negligible, as it typically remains relatively stable (22). During the analysis, we calculated the $k_{\rm sol}$ for solubility, which was approximately 0.9903 (±0.0091) during 2007–2020 (Fig. 4A). This result indicates that solubility rarely exhibited lower values during the day compared with nighttime; however, the difference was so minimal that it could be safely disregarded.

Following that, we examined the impacts of both the gas transport rate and the air–sea partial pressure difference on the diurnal flux variation, quantified by $k_{\Delta p}$ and k_{kw} , respectively. The diurnal partial pressure difference is the only parameter that can yield

negative values in the flux estimation. However, only 7% of them were negative, which indicated that diurnal variation did not change the source–sink properties of the oceans in most cases. However, the absolute average value of $k_{\Delta p}$ was approximately 0.98, ranging from 0.76 to 1.10. The distribution of $k_{\Delta p}$ demonstrated that the partial pressure difference influenced the magnitude of the flux, but this effect was relatively neutral and did not drive the flux to be higher at night than during the day.

The air–sea gas transfer velocity and the square of wind speed were positively correlated with diurnal wind speed. As shown in Fig. 4B, the value of k_{kw} was approximately 0.86 (±0.61), meaning that the larger nighttime wind speed caused a larger gas transfer rate at night. The ratio of $k_{kw}/|k_{\Delta p}|$ ranged from 0.8 to 5, implying that the contribution of wind speed and partial pressure difference to the effect of diurnal flux difference was approximately 7:1. In conclusion, the difference in diurnal air–sea C-flux was mainly controlled by the gas exchange rate affected by the wind speed, while the partial pressure difference determined the diffusion direction of air–sea CO₂.

Figure 3 reveals that the nocturnal C-flux is larger than during the daytime, influenced by a combination of diffusion, and biological factors, including photosynthesis and respiration. The powerful thermodynamic influence on seawater pCO₂ is evident, as higher daytime temperatures causing an impact on the carbonate balance and solubility, increasing more in the daytime. However, the biological effect is precisely the opposite, indicating higher nocturnal pCO₂ due to the respiration. These two effects determine the direction of diurnal pCO₂ changes. When thermodynamic photosynthesis is powerful, high biomass usually causes obvious respiration at night, resulting in the pCO_2 is larger during the night (Fig. 3A), and vice versa (Fig. 3B). The spatial distribution map of diurnal pCO₂ is shown in Fig. 3C, where both higher nighttime pCO₂ and diurnal pCO₂ are present. Despite the differences in diurnal pCO₂, ocean pCO₂ is lower than atmospheric pCO₂ globally, resulting in seawater being a sink for the atmosphere regardless of day and night. Additionally, the C-flux is affected by diffusion, which makes the wind speed determine the gas exchange rate. As a result, higher wind speeds at night accelerate the absorption of atmospheric CO_2 at night (Fig. 4A and B). The air-sea flux represents a dynamic steady state that depends on both pCO₂ and the air-sea turbulence exchange velocity. Colder surface water at night over a clear sky due to long wave radiation can help push more CO_2 into water in two ways: (i) colder water allows more gases in the water than warmer water; and (ii) mixed layer is more unstable at night and thus helps increase turbulence exchange velocity, when the drop in surface water temperature leads to heavier water at the surface. Furthermore, in the combined of all effects, the gas transfer rate flux has the greatest impact as mentioned in Fig. 4B. Hence, the global annual average C-flux is larger than at daytime under the multieffects.

In this study, we constructed a new FNN-LID method including CALIOP data for global pCO_2 and formed a dataset of long time-series variations in pCO_2 and air-sea CO_2 fluxes during 1998–2020, as well as diurnal products from 2007 to 2020. Instead of normal passive remote sensing products, we added CALIOP diurnal Chla and wind speed as the input data to build the FNN-LID approach. This is the first time LiDAR has been used in research on the air-sea carbon cycle. CALIOP can provide not only a larger coverage, including the polar regions, but also a clearer understanding of the global diurnal pCO_2 variations.

Globally, our FNN-LID demonstrated excellent fitting capabilities with the gridded pCO_2 data from in situ SOCATv2022, achieving an r^2 value of 0.79 and a negligible overall bias of 0.05 µatm over the entire time period from 1998 to 2020. These results showcased



Fig. 4. Histograms of diurnal ratios for each parameter: A) k_{sol} . B) k_{kw} , and C) $k_{\Delta p}$ involved in CO₂ flux calculations. Schematic representation of the comprehensive effects on diurnal C-fluxes in two different situations: D) biological effect larger than thermodynamic effect and E) biological effect lower than thermodynamic effect. Colorful arrows represent the effect of different influences on pCO_2 , and black ones for the results of synthetic effects. The area of the cylinder represents the amount of CO₂ flux, and the height of the lines represents the value of pCO_2 .

a similar spatiotemporal distribution and accuracy to those of other recent models in the mean monthly result. In addition, we found that pCO₂ exhibited different spatial characteristics during the day and at night by analyzing the FNN-LID results. It exhibited a unique spatial distribution, showing positive dpCO₂ in the subequatorial and midlatitude regions (higher diurnal pCO₂), while it was negative in the equatorial Pacific and southeast Pacific (higher nocturnal pCO_2), and the lowest $dpCO_2$ (more than 10 μ atm) was exhibited in Patagonia. The nocturnal CO₂ sink was higher than the diurnal sink, leasing to a potential underestimation between -0.005 and 0.052 GtC year⁻¹ if the diurnal uptake was used as the whole daily sink. This may be helpful in answering the question of where part of the carbon budget imbalance (approximately 0.1 GtC year⁻¹) came from over the last decade. Meanwhile, the contribution of wind speed and partial pressure difference to the effect of diurnal flux difference is approximately 7:1. The partial pressure difference played a crucial role in determining the diffusion direction of air-sea CO2 exchange.

This model's application has extended the use of satellite remote sensing data in polar research and diurnal variation studies. Moving forward, with the accumulation of remote sensing data, we will further extend the study of medium time scale carbonate systems. Additionally, we will explore the utilization of higher spatial and temporal resolution data to analyze carbonate systems in coastal and nearshore ecosystems, further enhancing our understanding of these vital marine environments.

Materials and methods

Observation data

The gridded monthly pCO_2 data were sourced from the gridded SOCATv2022 observational database (available at https://www.socat.info/) (83). SOCATv2022 provides comprehensive global sea surface fugacity of CO_2 (fCO_2) data from moorings, ships, and drifters spanning the period from 1970 to 2020. Furthermore, wind speed data from the tropical moored buoys, along with b_{bp} and Chla products from an array of Bio-Argo floats, were utilized to validate the accuracy of CALIOP Chla, and the wind speed. Refer to SI Appendix for preprocessing of observation measurements in this study.

CALIOP measurements

The active remote sensing data came from CALIOP, developed by NASA (available at http://orca.science.oregonstate.edu/lidar.

data.php), including CALIPSO level 1B V4.10 data products, LiDAR level 2 Cloud, Aerosol, and Merged Layer V4.20 products. The measured signal is corrected for after-pulse and polarization crosstalk effects (84) before further processed. Following these corrections, $b_{\rm bp}$ can be calculated from the vertical-parallel ratio (34, 85). Finally, Chla can be estimated based on the relational formula of $b_{\rm bp}$ and chlorophyll-a concentration C.

Ocean surface roughness increases with higher wind speeds, resulting in fewer photons reaching the LiDAR receiver. Thus, the LiDAR backscatter signal is proportionally related to the probability that the surface of the capillary-gravity waves is perpendicular to the line-of-sight of the laser beam (86). In this study, wind speed data were retrieved from LiDAR data from Version 4 CALIPSO level 1 data in combination with collocated ocean surface wind speed data from AMSR-E measurements (87). Refer to SI Appendix for further information on the method and the fitting neural network (SI Appendix, Table S4).

Satellite and reanalyzed environmental datasets

As shown in SI Appendix, Table S5, our predictors included biological, chemical, and physical variables commonly associated with variations in pCO_2 (13, 17, 88). These predictors comprised SST, sea surface salinity, sea surface height, mixed-layer depth, Chla, and atmospheric CO_2 mole fraction (xCO_2). It is noteworthy to mention that the Chla used in this study is based on optical measurements rather than analyzed Chla data. In addition to the predictors listed in this table, pCO₂ climatology (89), normalized latitude, and longitude were also utilized as predictors for the reconstruction. Furthermore, the calculation of the C-flux requires additional datasets, including atmospheric pressure, and 10 m wind speed at the sea surface. The original data were distributed after interpolation on 1° latitude by 1° longitude cells. For reconstructing the diurnal pCO₂ field and calculating the air-sea flux, we used all data capable of diurnal spatiotemporal resolution, which encompassed Chla, wind speed, temperature, barometric pressure, and atmospheric CO₂ concentration. In diurnal research, these data were considered to have the same spatial resolution and began in 2007. See SI Appendix for information on the satellite and reanalyzed environmental datasets used in this study.

Reconstruction datasets

In the validation of the results, we also employed several widely used pCO_2 reconstruction datasets to corroborate our findings, including pCO_2 data reconstructed using FNN from CMEMS and pCO_2 data reconstructed using SOM-FNN from IBP. These datasets share the same global monthly long-term time series (from 1998 to 2020) at a resolution of 1°×1°.

FNN-LID model

It is very difficult (probably impossible, or hopeless) to measure ocean surface pCO_2 directly from space. Ocean surface pCO_2 is often constructed based on global ocean biogeochemical models and data reconstruction methods with satellite remote sensing environmental data, SI Appendix, Fig. S3D. In this study, we adopted a novel feed-forward neural network methods including CALIOP data to reconstruct the diurnal pCO_2 for the period from January 1998 to December 2020 at a monthly resolution of $1^\circ \times 1^\circ$. The two-part method utilized here established nonlinear relationships between pCO_2 and a set of independent environmental predictors. In the first part, we derived a nonlinear and continuous relationship between climatology pCO_2 , and the independent environmental predictors based on an FNN method. For the second

part, we used the gridded SOCATv2022 observational database as the target.

On this basis, we updated the input data for the diurnal CALIPSO surface sea Chla, diurnal xCO_2 from ECMWF and the diurnal SST product from MODIS during 2007–2020. Hence, we reconstructed the day and night sea surface partial pressure of CO_2 for the period from January 2007 to December 2021 at a monthly resolution of 1°×1°. For further details on the FNN-LID model, refer to SI Appendix.

Acknowledgments

The authors are grateful to the NASA Langley Research Center for providing CALIOP data through the Atmospheric Sciences Data Center (https://asdc.larc.nasa.gov/data/CALIPSO/). The authors thank the NASA Marine Biology Processing Group and Globe Color for also providing MODIS products (https://oceandata.sci. gsfc.nasa.gov/). They also thank anonymous reviewers for their suggestions, which significantly improved the presentation of the paper.

Supplementary Material

Supplementary material is available at PNAS Nexus online.

Funding

This work was supported by the National Natural Science Foundation (42322606; 42276180; 61991453), the National Key Research and Development Program of China (2022YFB3901703; 2022YFB3902603), the Key Special Project for Introduced Talents Team of Southern Marine Science and Engineering Guangdong Laboratory (GML2021GD0809), the Donghai Laboratory Preresearch project (DH2022ZY0003), and the Key Research and Development Program of Zhejiang Province (grant no. 2020C03100).

Author Contributions

S.Z.: writing—original draft preparation. P.C.: conceptualization, methodology, writing—original draft preparation. Z.Z.: software, visualization. Y.H. X.L.: data, methodology, writing—review and editing. C.J. and D.D.: validation, methodology, writing—review and editing, formal analysis. D.P.: supervision, project administration, investigation, and funding acquisition.

Data Availability

All LiDAR data are publicly available at https://zenodo.org/record/7047301. All pCO_2 and C-flux data are included in supplementary material.

References

- 1 Weiman S. 2015. Microbes help to drive global carbon cycling and climate change. *Microbe Mag.* 10:233–238.
- 2 McNeil BI, Sasse TP. 2016. Future ocean hypercapnia driven by anthropogenic amplification of the natural CO₂ cycle. Nature. 529:383–386.
- 3 Friedlingstein P, et al. 2022. Global carbon budget 2021. Earth Syst Sci Data. 14:1917–2005.
- 4 Friedlingstein P, et al. 2020. Global carbon budget 2020. Earth Syst Sci Data Discuss. 12:3269–3340.

- 5 Friedlingstein P, et al. 2022. Global carbon budget 2022. Earth Syst Sci Data Discuss. 14:4811–4900.
- 6 Fay AR, et al. 2021. Harmonization of global surface ocean pCO₂ mapped products and their flux calculations; an improved estimate of the ocean carbon sink. Earth Syst Sci Data Discuss. 13(10):4693–4710.
- 7 Rödenbeck C, et al. 2015. Data-based estimates of the ocean carbon sink variability-first results of the Surface Ocean pCO₂ Mapping intercomparison (SOCOM). Biogeosciences. 12:7251–7278.
- 8 Le Quéré C, et al. 2018. Global carbon budget 2017. Earth Syst Sci Data. 10:405–448.
- 9 Rödenbeck C, *et al.* 2014. Interannual sea–air CO flux variability from an observation-driven ocean mixed-layer scheme. *Biogeosciences.* 11:4599–4613.
- 10 Lefèvre N, Watson AJ, Watson AR. 2005. A comparison of multiple regression and neural network techniques for mapping in situ pCO_2 data. Tellus B: Chem Phys Meteorol. 57:375–384.
- 11 Friedrich T, Oschlies A. 2009. Neural network-based estimates of North Atlantic surface pCO₂ from satellite data: a methodological study. J Geophys Res: Oceans. 114:1–12.
- 12 Telszewski M, et al. 2009. Estimating the monthly pCO₂ distribution in the North Atlantic using a self-organizing neural network. *Biogeosciences.* 6:1405–1421.
- 13 Landschützer P, et al. 2013. A neural network-based estimate of the seasonal to inter-annual variability of the Atlantic Ocean carbon sink. Biogeosciences. 10:7793–7815.
- 14 Nakaoka S-i, et al. 2013. Estimating temporal and spatial variation of ocean surface pCO_2 in the North Pacific using a selforganizing map neural network technique. *Biogeosciences*. 10: 6093–6106.
- 15 Zeng J, Nojiri Y, Landschützer P, Telszewski M, Nakaoka S-I. 2014. A global surface ocean fCO₂ climatology based on a feed-forward neural network. J Atmos Ocean Technol. 31:1838–1849.
- 16 Laruelle GG, et al. 2017. Global high-resolution monthly pCO₂ climatology for the coastal ocean derived from neural network interpolation. *Biogeosciences*. 14:4545–4561.
- 17 Denvil-Sommer A, Gehlen M, Vrac M, Mejia C. 2019. FFNN-LSCE: a two-step neural network model for the reconstruction of surface ocean pco₂ over the global ocean. *Geosci Model Dev.* 12: 2091–2105.
- 18 Chau TTT, Gehlen M, Chevallier F. 2022. A seamless ensemblebased reconstruction of surface ocean pCO_2 and air-sea CO_2 fluxes over the global coastal and open oceans. *Biogeosciences*. 19:1087–1109.
- 19 Burger FA, John JG, Frölicher TL. 2020. Increase in ocean acidity variability and extremes under increasing atmospheric CO₂. Biogeosciences. 17:4633–4662.
- 20 Torres O, Kwiatkowski L, Sutton AJ, Dorey N, Orr JC. 2021. Characterizing mean and extreme diurnal variability of ocean CO₂ system variables across marine environments. *Geophys Res* Lett. 48:e2020GL090228.
- 21 Bates NR, Takahashi T, Chipman DW, Knap AH. 1998. Variability of pCO₂ on diel to seasonal timescales in the Sargasso Sea near Bermuda. J Geophys Res: Oceans. 103:15567–15585.
- 22 Olsen A. 2004. Diurnal variations of surface ocean pCO_2 and sea-air CO_2 flux evaluated using remotely sensed data. Geophys Res Lett. 31:1–4.
- 23 Drupp PS, et al. 2013. Comparison of CO₂ dynamics and air-sea gas exchange in differing tropical reef environments. Aquat Geochem. 19:371–397.
- 24 Albright R, et al. 2018. Carbon dioxide addition to coral reef waters suppresses net community calcification. Nature. 555: 516–519.

- 25 Murie KA, Bourdeau PE. 2020. Fragmented kelp forest canopies retain their ability to alter local seawater chemistry. Sci Rep. 10: 11939.
- 26 Berg P, et al. 2019. Dynamics of benthic metabolism, O₂, and pCO₂ in a temperate seagrass meadow. Limnol Oceanogr. 64:2586–2604.
- 27 Hofmann GE, et al. 2011. High-frequency dynamics of ocean pH: a multi-ecosystem comparison. PLoS One. 6:e28983.
- 28 Page HN, et al. 2019. Spatiotemporal variability in seawater carbon chemistry for a coral reef flat in Kāne 'ohe Bay, Hawai 'i. Limnol Oceanogr. 64:913–934.
- 29 Jury CP, et al. 2013. Buffer capacity, ecosystem feedbacks, and seawater chemistry under global change. Water. 5:1303–1325.
- 30 Denman K, Christian JR, Steiner N, Pörtner H-O, Nojiri Y. 2011. Potential impacts of future ocean acidification on marine ecosystems and fisheries: current knowledge and recommendations for future research. ICES J Mar Sci. 68:1019–1029.
- 31 Kranz SA, Dieter S, Richter K-U, Rost B. 2009. Carbon acquisition by Trichodesmium: the effect of pCO_2 and diurnal changes. Limnol Oceanogr. 54:548–559.
- 32 Peter H, et al. 2014. Scales and drivers of temporal pCO₂ dynamics in an Alpine stream. J Geophys Res: Biogeosci. 119:1078–1091.
- 33 Winker D, et al. 2010. The CALIPSO mission: a global 3D view of aerosols and clouds. Bull Am Meteorol Soc. 91:1211–1230.
- 34 Behrenfeld MJ, et al. 2013. Space-based lidar measurements of global ocean carbon stocks. Geophys Res Lett. 40:4355–4360.
- 35 Lu X, Hu Y, Trepte C, Zeng S, Churnside JH. 2014. Ocean subsurface studies with the CALIPSO spaceborne lidar. J Geophys Res: Oceans. 119:4305–4317.
- 36 Churnside JH, McCarty BJ, Lu X. 2013. Subsurface ocean signals from an orbiting polarization lidar. *Remote Sens.* 5:3457–3475.
- 37 Behrenfeld MJ, et al. 2017. Annual boom-bust cycles of polar phytoplankton biomass revealed by space-based lidar. Nat Geosci. 10:118–122.
- 38 Bisson K, Boss E, Werdell P, Ibrahim A, Behrenfeld M. 2021. Particulate backscattering in the global ocean: a comparison of independent assessments. *Geophys Res Lett.* 48:e2020GL090909.
- 39 Hu Y, et al. 2008. Sea surface wind speed estimation from spacebased lidar measurements. Atmos Chem Phys. 8:3593–3601.
- 40 Hu Y, et al. 2009. CALIPSO/CALIOP cloud phase discrimination algorithm. J Atmos Oceanic Technol. 26:2293–2309.
- 41 Behrenfeld MJ, et al. 2019. Global satellite-observed daily vertical migrations of ocean animals. Nature. 576:257–261.
- 42 Lu X, et al. 2020. Antarctic spring ice-edge blooms observed from space by ICESat-2. Remote Sens Environ. 245:111827.
- 43 Dionisi D, Brando VE, Volpe G, Colella S, Santoleri R. 2020. Seasonal distributions of ocean particulate optical properties from spaceborne lidar measurements in Mediterranean and Black Sea. *Remote Sens Environ.* 247:111889.
- 44 Jamet C, et al. 2019. Going beyond standard ocean color observations: lidar and polarimetry. Front Mar Sci. 6:251.
- 45 Hostetler CA, Behrenfeld MJ, Hu Y, Hair JW, Schulien JA. 2018. Spaceborne lidar in the study of marine systems. Annu Rev Mar Sci. 10:121–147.
- 46 Claustre H, Johnson KS, Takeshita Y. 2019. Observing the global ocean with biogeochemical-argo. Annu Rev Mar Sci. 12:1–22.
- 47 Bakker DCE, et al. 2016. A multi-decade record of high-quality fCO₂ data in version 3 of the Surface Ocean CO₂ Atlas (SOCAT). Earth Syst Sci Data. 8:383–413.
- 48 Landschützer P, Gruber N, Bakker DC, Schuster U. 2014. Recent variability of the global ocean carbon sink. Glob Biogeochem Cycles. 28:927–949.
- 49 Hales B, et al. 2012. Satellite-based prediction of pCO₂ in coastal waters of the eastern North Pacific. Prog Oceanogr. 103:1–15.

- 50 Zhang S, Chen P, Zhang Z, Pan D. 2022. Carbon air–sea flux in the Arctic ocean from CALIPSO from 2007 to 2020. *Remote Sens.* 14: 6196.
- 51 Babin M, Arrigo K, Bélanger S, Forget M-H, International Ocean Colour Coordinating Group. 2015. Ocean colour remote sensing in polar seas. 51–52.
- 52 Liu Y, Ackerman SA, Maddux BC, Key JR, Frey RA. 2010. Errors in cloud detection over the Arctic using a satellite imager and implications for observing feedback mechanisms. J Clim. 23: 1894–1907.
- 53 Betts RA, et al. 2020. ENSO and the carbon cycle. El Niño Southern Oscillat Change Clim. 4:453–470.
- 54 Amante C, Eakins BW, NATIONAL OCEANIC AND ATMOSPHERIC ADMINISTRATION National Environmental Satellite, Data, and Information Service. 2009. ETOPO1 arc-minute global relief model: procedures, data sources and analysis. 1–3.
- 55 Eakins B, Sharman G. 2010. Volumes of the world's oceans from ETOPO1. Boulder (CO): National Oceanic and Atmospheric Administration National Geophysical Data Center.
- 56 Jacobson AR, Mikaloff Fletcher SE, Gruber N, Sarmiento JL, Gloor M. 2007. A joint atmosphere-ocean inversion for surface fluxes of carbon dioxide: 1. Methods and global-scale fluxes. Glob Biogeochem Cycles. 21:1–13.
- 57 Resplandy L, et al. 2018. Revision of global carbon fluxes based on a reassessment of oceanic and riverine carbon transport. Nat Geosci. 11:504–509.
- 58 Takahashi T, Olafsson J, Goddard JG, Chipman DW, Sutherland S. 1993. Seasonal variation of CO₂ and nutrients in the high—latitude surface oceans: a comparative study. *Glob Biogeochem* Cycles. 7:843–878.
- 59 Fairall C, et al. 1996. Cool-skin and warm-layer effects on sea surface temperature. J Geophys Res: Oceans. 101:1295–1308.
- 60 Clayson C. 2007. Effects of a diurnal sea surface temperature on surface fluxes and atmospheric variability. In: AMS 19th Conference on Climate Variability and Change, San Diego, CA, USA, p. 15–18.
- 61 Prézelin BB. 1992. The daily growth cycle of phytoplankton. Berlin: Springer. p. 1–35.
- 62 Vaulot D, Marie D, Olson RJ, Chisholm SW. 1995. Growth of Prochlorococcus, a photosynthetic prokaryote, in the equatorial Pacific Ocean. *Science*. 268:1480–1482.
- 63 Liu H, Nolla HA, Campbell L. 1997. Prochlorococcus growth rate and contribution to primary production in the equatorial and subtropical North Pacific Ocean. Aquat Microb Ecol. 12:39–47.
- 64 Vaulot D, Marie D. 1999. Diel variability of photosynthetic picoplankton in the equatorial Pacific. J Geophys Res: Oceans. 104: 3297–3310.
- 65 Binder BJ, DuRand MD. 2002. Diel cycles in surface waters of the equatorial Pacific. *Deep Sea Res Part II: Top Stud Oceanogr.* 49: 2601–2617.
- 66 Ng WHA, Liu H. 2015. Diel variation of the cellular carbon to nitrogen ratio of Chlorella autotrophica (Chlorophyta) growing in phosphorus—and nitrogen—limited continuous cultures. J Phycol. 51:82–92.
- 67 Delille B, Borges AV, Delille D. 2009. Influence of giant kelp beds (Macrocystis pyrifera) on diel cycles of pCO2 and DIC in the Sub-Antarctic coastal area. Estuar Coast Shelf Sci. 81:114–122.
- 68 Dore JE, Lukas R, Sadler DW, Church MJ, Karl DM. 2009. Physical and biogeochemical modulation of ocean acidification in the central North Pacific. Proc Natl Acad Sci U S A. 106:12235–12240.

- 69 Gruber N, Keeling CD, Bates NR. 2002. Interannual variability in the North Atlantic Ocean carbon sink. Science. 298:2374–2378.
- 70 Bates N. 2012. Multi-decadal uptake of carbon dioxide into subtropical mode water of the North Atlantic Ocean. *Biogeosciences*. 9:2649–2659.
- 71 Phillips HE, Joyce TM. 2007. Bermuda's tale of two time series: hydrostation S and BATS. J Phys Oceanogr. 37:554–571.
- 72 Doney SC, Fabry VJ, Feely RA, Kleypas JA. 2009. Ocean acidification: the other CO₂ problem. Annu Rev Mar Sci. 1:169–192.
- 73 Hauck J, Völker C. 2015. Rising atmospheric CO₂ leads to large impact of biology on Southern Ocean CO₂ uptake via changes of the Revelle factor. *Geophys Res Lett.* 42:1459–1464.
- 74 Walton C, Pichel W, Sapper J, May D. 1998. The development and operational application of nonlinear algorithms for the measurement of sea surface temperatures with the NOAA polar-orbiting environmental satellites. J Geophys Res: Oceans. 103:27999–28012.
- 75 Kilpatrick KA, et al. 2015. A decade of sea surface temperature from MODIS. *Remote Sens Environ*. 165:27–41.
- 76 Donlon C, et al. 2007. The global ocean data assimilation experiment high-resolution sea surface temperature pilot project. Bull Am Meteorol Soc. 88:1197–1214.
- 77 Hanafin JA, University of Miami ProQuest Dissertations Publishing. 2002. On sea surface properties and characteristics in the infrared.
- 78 Wong EW, Minnett PJ. 2018. The response of the ocean thermal skin layer to variations in incident infrared radiation. J Geophys Res: Oceans. 123:2475–2493.
- 79 Corlett GK, Merchant CJ, Minnett PJ, Donlon CJ, Elsevier. 2014. Experimental methods in the physical sciences. Giuseppe Zibordi. Amsterdam: Elsevier. p. 639–67747.
- 80 Lu X, et al. 2016. Retrieval of ocean subsurface particulate backscattering coefficient from space-borne CALIOP lidar measurements. Opt Express. 24:29001–29008.
- 81 Bisson K, et al. 2021. Seasonal bias in global ocean color observations. Appl Opt. 60:6978–6988.
- 82 Roesler C, et al. 2017. Recommendations for obtaining unbiased chlorophyll estimates from in situ chlorophyll fluorometers: a global analysis of WET Labs ECO sensors. *Limnol Oceanogr:* Methods. 15:572–585.
- 83 Bakker DC, et al. 2016. A multi-decade record of high-quality fCO₂ data in version 3 of the Surface Ocean CO₂ Atlas (SOCAT). Earth Syst Sci Data. 8:383–413.
- 84 Lu X, Hu Y, Vaughan M, Rodier S, Omar A. 2020. New attenuated backscatter profile by removing the CALIOP receiver's transient response. J Quant Spectrosc Radiat Transf. 255:107244.
- 85 Bisson KM, Boss E, Werdell PJ, Ibrahim A, Behrenfeld MJ. 2021. Particulate backscattering in the global ocean: a comparison of independent assessments. *Geophys Res Lett.* 48:e2020GL090909.
- 86 Murphy A, Hu Y. 2021. Retrieving aerosol optical depth and high spatial resolution ocean surface wind speed from CALIPSO: a neural network approach. Front Remote Sens. 1:614029.
- 87 Vaughan M, et al. 2019. CALIPSO lidar calibration at 1064 nm: version 4 algorithm. Atmos Meas Tech. 12:51–82.
- 88 Gregor L, Lebehot AD, Kok S, Scheel Monteiro PM. 2019. A comparative assessment of the uncertainties of global surface ocean CO₂ estimates using a machine-learning ensemble (CSIR-ML6 version 2019a)–have we hit the wall? *Geosci Model Dev.* 12:5113–5136.
- 89 Takahashi T, et al. 2009. Climatological mean and decadal change in surface ocean pCO₂, and net sea–air CO₂ flux over the global oceans. Deep Sea Res Part II: Top Stud Oceanogr. 56:554–577.